Author’s response to reviews

Title: Electronic self-monitoring of mood using IT platforms in adult patients with bipolar disorder: A systematic review of the validity and evidence

Authors:

Maria Faurholt-Jepsen (maria@faurholt-jepsen.dk)
Klaus Munkholm (klaus.munkholm@regionh.dk)
Mads Frost (madsf@itu.dk)
Jakob Bardram (bardram@itu.dk)
Lars Kessing (lars.vedel.kessing@regionh.dk)

Version: 4 Date: 20 Nov 2015

Author’s response to reviews:

Cover Letter

BMC Psychiatry Copenhagen,
November 2015

Dear Iskra Staneva,

Please find the revision of the systematic review article:

“Electronic self-monitoring of mood using IT platforms in adult patients with bipolar disorder: A systematic review of the validity and evidence”

We have addressed all of your editorial comments and revised the manuscript accordingly. With this revision, we hope that you will consider publishing our manuscript in BMC Psychiatry.

Listed below are the comments along with our reply in a point-by-point manner.
Comments:

1) Comment: Ethics issues should be addressed.
Reply: A section on ethics has been provided in the Methods section.

2) Comment: Consent issues should be addressed.
Reply: A section on consent has been provided in the ethical part of the Methods section.

3) Comment: Availability of data
Reply: Since this is a systematic review all data included in the paper is published and therefore available.

4) Comment: Authors Contributions should be addressed.
Reply: The section on Contributions has been revised and includes more details.

Sincerely yours,

Maria Faurholt-Jepsen & Lars Vedel Kessing
MD Professor, DMSc
Psychiatric Center Copenhagen, Rigshospitalet
University of Copenhagen
Denmark